Author's response to reviews

Title: The Cloud Paradigm Applied to e-Health

Authors:

Jordi Vilaplana (jordi@diei.udl.cat)
Francesc Solsona (francesc@diei.udl.cat)
Francesc Abella (abella@gss.scs.es)
Rosa Filgueira (rosa.filgueira@ed.ac.uk)
Josep Rius (jrius@diei.udl.cat)

Version: 5 Date: 26 January 2013

Author's response to reviews: see over
Abstract

Point-by-point response to the reviewer's comments.

1 Reviewer 1

1.1 Comment 1

“The authors may refer to some published articles to strengthen their argument about the contribution of the study (i.e. cloud computing based platform) to e-Healthcare. For example, Alex M. H. Kuo, “Opportunities and Challenges of Cloud Computing to Improve Health Care Services,” Journal of Medical Internet Research (JMIR), 13(3):e67, 2011.”

We have added more references and we have reviewed all the manuscript.

1.1.1 Solution

The following references have been added to the Background section of the manuscript.

Cloud computing can offer many opportunities to improve health care services from the viewpoint of management, technology, security and legality [1].

In [2], the authors examine how the biomedical informatics community, especially consortia that share data and applications, can take advantage of cloud computing.

References
